Madison, Wilton

18-01776-E

From:

Mark Edwards < medwards@biosciadvisors.com>

Sent:

Saturday, January 06, 2018 7:33 PM

To:

foiapa

Subject:

FOIA Request

I would like to request access to Exhibit 10.2 to the 9/30/12 10-Q, filed by ViroPharma, Inc. on 10/25/2012. Confidential treatment was sought as to certain portions when initially filed with the Commission.

In the event that confidential treatment has not expired or has been extended, I further request that you send me the expiration date(s) from the relevant CT order(s) so I will know when I should resubmit my request.

I authorize up to \$61 in search and retrieval fees. Please send the exhibit(s) by PDF if possible.

Sincerely,

Mark

Mark G Edwards
Managing Director
Bioscience Advisors
2855 Mitchell Dr., Suite 103
Walnut Creek, CA 94598
medwards@biosciadvisors.com
925 954-1397

RECEIVED

JAN 08 2018

Office of FOIA Services



UNITED STATES SECURITIES AND EXCHANGE COMMISSION

STATION PLACE 100 F STREET, NE WASHINGTON, DC 20549-2465

Office of FOIA Services

January 30, 2018

Mr. Mark G. Edwards Bioscience Advisors 2855 Mitchell Dr., Suite 103 Walnut Creek, CA 94598

RE: Freedom of Information Act (FOIA), 5 U.S.C. § 552

Request No. 18-01776-E

Dear Mr. Edwards:

This letter is in response to your request, dated January 06, 2018 and received in this office on January 08, 2018, for information regarding Exhibit 10.2 to the Form 10-Q (dated September 30, 2012), filed by ViroPharma, Inc. on October 25, 2012.

The search for responsive records has resulted in the retrieval of 42 pages that may be responsive to your request. They are being provided to you with this letter in their entirety at no cost.

If you have any questions, please contact me at Luetkenhausj@SEC.GOV or (202) 551-8352. You may also contact me at foiapa@sec.gov or (202) 551-7900. You also have the right to seek assistance from Aaron Taylor as a FOIA Public Liaison or contact the Office of Government Information Services (OGIS) for dispute resolution services. OGIS can be reached at 1-877-684-6448 or Archives.gov or via e-mail at ogis@nara.gov.

Sincerely,

Jason Luetkenhaus Lead FOIA Research Specialist

Enclosures

10.2

2012 STRATEGIC SUPPLY AGREEMENT

This Strategic Supply Agreement (this "Agreement") is entered into as of the 24th day of September 2012 (the "Effective Date"), by and between ViroPharma Biologics, Inc., a Delaware corporation, having an address of 730 Stockton Drive, Exton, PA 19341 (the "Buyer") and Biotest Pharmaceuticals Corporation, a Delaware corporation, having an address of 5800 Park of Commerce Boulevard NW, Boca Raton, FL 33487 ("Seller"). Buyer and Seller are referred to collectively herein as the "Parties" and each individually as a "Party."

RECITALS

WHEREAS, on February 26, 2010, the Parties entered into a Strategic Supply Agreement ("2010 Agreement"), whereby the Seller would construct and operate three (3) plasma collection centers, with the option to construct and operate up to two (2) additional plasma collection centers in the United States; and

WHEREAS, in addition, pursuant to the 2010 Agreement, Seller is collecting and supplying Plasma to Buyer, and Buyer is purchasing such Plasma from Seller; and

WHEREAS, further under the 2010 Agreement, Buyer has the option to purchase the plasma collection centers and such related assets, properties and rights from the Seller; and

WHEREAS, Buyer provided notice to the Seller on June 14, 2012 that it wishes Seller to construct and operate four (4) additional plasma centers, rather than the two (2) additional plasma collection centers as originally set forth in the 2010 Agreement; and

WHEREAS, Seller and Buyer therefore wish to terminate the 2010 Agreement as of the Effective Date of this Agreement and enter into a new agreement whereby the Seller would construct and operate an additional four (4) plasma collection centers and Seller would continue to supply Plasma to Buyer,

NOW, THEREFORE, in consideration of the premises and the respective promises herein made, and in consideration of the representations, warranties, and covenants herein contained, the Parties agree as follows:

Section 1. Certain Definitions.

"Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly controlling, controlled by, or under common control with such Person. As used in this definition, the term "control" (including the terms "controlling," "controlled by" and "under common control with") means possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of voting securities, by contract or otherwise.

"<u>Alternative Supplier</u>" shall mean a supplier of Plasma that wishes to supply Plasma to the Buyer, other than Seller.

"Applicable Laws" shall mean all applicable federal, state, local and foreign laws, requirements, regulations, guidelines, licenses and directives, including the Federal Food, Drug and Cosmetic Act and applicable current Good Manufacturing Practices ("cGMP"), including all cGMP for blood and blood components, specifications and procedures for plasma sourcing, plasma testing, and in process testing and all regulations, specifications, and procedures contained therein, including such laws, requirements, regulations, guidelines, licenses and directives promulgated by the United States Food and Drug Administration (or any successor agency thereto) ("FDA") or the European Medicines Agency (or any successor thereto) ("EMEA").

"Approved Testing Centers" shall mean those facilities (as set forth in Schedule "B") that will test the Plasma, to be provided under this Agreement, in accordance with FDA and EMEA requirements and the Specifications.

"BPC Centers" shall mean those plasma collection centers that Seller owns and operates at various sites throughout the United States other than the ViroPharma Centers.

"Buyer Approved Collection Center" shall have the meaning provided in Section 4.6.

"<u>CPI</u>" shall mean the increase (if any) in the Consumer Price Index published by the Bureau of Labor Statistics of the United States Department of Labor for all Urban Consumers (CPI-U) - - U.S. All Items (1982-1984 = 100) for the prior November 1-October 31 period.

"Center Pre-Payment" shall mean the sum of Seven Hundred Fifty Thousand Dollars (\$750,000) which is payable as set forth in Section 3.2.2.

"Costs" shall mean Seller's fully loaded costs to include all direct, indirect and fixed manufacturing costs and Seller's allocated Plasma Operations and General & Administrative costs.

"<u>Development Proposal</u>" shall mean a proposal that includes the proposed location, terms of the lease, floor plan and other reasonable information regarding the development of each New Center.

"Foreign Regulatory Approvals" shall mean all necessary regulatory approvals, permits, certifications, consents, licenses, registrations, listings, certificates of origin, and any other requisite documents and authorizations for the operation of a Plasma collection facility and the sourcing, collection, manufacture, testing, marketing, use, sale, handling, storage or distribution of Plasma, pursuant to any laws, treaties, statues, regulations or other requirements applicable in the European Union or any member country of the European Union.

"Initial Centers" shall mean three (3) initial plasma collection centers that have been developed and opened pursuant to the terms of the 2010 Agreement, as more fully set forth on Exhibit "C".

"Initial Term" shall mean the period of time from the Effective Date of this Agreement until December 31, 2017.

"<u>Management Agreement</u>" shall mean the definitive agreement which sets forth the terms and conditions whereby the Seller will manage the ViroPharma Centers in the event that the ViroPharma Centers are purchased by the Buyer.

"Material Adverse Effect" shall mean any effect or change that would be materially adverse to the business, assets, condition (financial or otherwise), operating results, or operations, of the BPC Centers, ViroPharma Centers or Seller or to the ability of any Party to consummate timely the transactions contemplated hereby.

"New Center(s)" shall mean those four (4) additional plasma collection centers in the Eastern, Central and/or Mountain Time zones in the United States.

"New Center Development Payment" shall mean the two (2) non-refundable payments, in the amount of One Million (\$1,000,000) each to cover the start-up costs during the development phases of the New Centers, which are payable as set forth in Section 3.2.2.

"Non-Qualified Center" shall mean a New Center (defined above) that has not received all required Regulatory Approvals.

"Opening Date" shall mean the date on which a New Center commences business operations.

"Option Period" shall mean the period of time, as set forth in Section 2.5, during which Buyer may opt to purchase the ViroPharma Centers pursuant to the terms and conditions of this Agreement.

"Outstanding Pre-Payments" shall have the meaning ascribed to such term in Section 3.2.1.

"Person" shall mean any individual, corporation, partnership, limited liability company, association, trust or other entity or organization, including any governmental authority.

"Plasma" shall mean "Source Plasma" as defined by the FDA in 21 C.F.R. 640.60 that meets the definitions and specifications of Buyer set forth in Schedule 1 (as such specifications may be amended from time to time in accordance with the terms of this Agreement) (the "Specifications"), attached hereto and incorporated by reference herein.

"Plasma Pre-Payment" shall mean the sum of One Million Dollars (\$1,000,000) which is payable as set forth in Section 3.2.2.

"<u>Plasma Volume</u>" shall mean the annual calendar year commitment of Plasma that Seller agrees to sell and Buyer agrees to purchase as set forth in Sections 4.1 through 4.3 of the Agreement.

"Product" shall mean the human pharmaceutical product(s) processed or manufactured by or on behalf of Buyer using the Plasma purchased from Seller under this Agreement.

"Proposed Transaction" shall mean a transaction involving the acquisition of any capital stock, membership interests or other voting securities of either Party, by an entity in the plasma industry or which would result in a direct or indirect change in control of either Party (including by merger, consolidation, exchange, or purchase of a more than 50% controlling interest in either Party or any direct or indirect parent thereof) by an entity in the plasma industry or, in the case of Seller, a transaction involving the acquisition of all or substantially all of the Buyer Approved Collection Centers by an entity in the plasma industry.

"Purchase Agreement" shall mean the definitive asset purchase agreement that the Buyer and Seller shall negotiate in the event Buyer exercises its Purchase Option under Section 2.5 to require a sale and purchase of the ViroPharma Centers.

"<u>Purchase Option</u>" shall mean the Buyer's option to purchase the ViroPharma Centers, as set forth in section 2.5, pursuant to the terms and conditions of this Agreement.

"<u>Purchase Price</u>" shall mean the purchase price of the Plasma purchased by Buyer from the Seller, as set forth in Section 3.3(a) of this Agreement.

"Qualified Center" shall mean a New Center that has received all required Regulatory Approvals.

"Refundable Pre-Payments" shall mean the Plasma Pre-Payment and all Center Pre-Payments.

"Regulatory Approvals" shall mean all necessary regulatory approvals, permits, certifications, consents, licenses, registrations, listings, certificates of origin, and any other requisite documents and authorizations for the operation of a Plasma collection facility and the sourcing, collection, manufacture, testing, marketing, use, sale, handling, storage or distribution of Plasma, including FDA licensure, IQPP certification, CLIA and required state licensing, and including all Foreign Regulatory Approvals.

"Renewal Term" shall mean additional two (2) year periods, after the Initial Term.

"Term" shall mean the Initial Term and any Renewal Term(s) thereafter.

"<u>Total Pre-Payments</u>" shall mean the aggregate amount of all Outstanding Pre-Payments and Refundable Pre-Payments paid by Buyer to Seller.

"ViroPharma Centers" shall mean the Initial Centers and New Centers.

"<u>ViroPharma Center Assets</u>" shall mean those assets for each ViroPharma Center as set forth in a Purchase Agreement including without limitation the assets set forth in Section 5.6.

Section 2. General Terms and Obligations.

2.1 Supply of Plasma. During the Term and upon the terms and subject to the conditions of this Agreement, Buyer hereby agrees to purchase, and Seller agrees to supply, quantities of Plasma as further set forth herein.

- **2.2 Development of New Centers.** Pursuant to the terms and conditions of this Agreement, Seller shall design, construct and operate the ViroPharma Centers.
- 2.3 Development, Opening and Regulatory Approval for New Centers. Seller will use commercially reasonable efforts to develop, open and obtain Regulatory Approvals for the New Centers pursuant to the schedule below (the "Approval Deadline"):

| New Center | Development Proposal | Approval of Development Proposal by Buyer | Opening Date | Regulatory Approval Date |
|---------------|---|--|---|---|
| 1 | No later than 90 days from Effective Date | No later than 30 days from receipt of Development Proposal | No later than 10 months from Effective Date | No later than 22 months from Effective Date |
| 2 | No later than 8 months from Effective Date | No later than 30 days from receipt of Development Proposal | No later than 15 months from Effective Date | No later than 27 months from Effective Date |
| 3 | No later than 14 months from Effective Date | No later than 30 days from receipt of Development Proposal | No later than 21 months from Effective Date | No later than 33 months from Effective Date |
| 4 | No later than 20 months from Effective Date | No later than 30 days from receipt of Development Proposal | No later than 27 months from Effective Date | No later than 39 months from Effective Date |

provided, however, that, except as otherwise set forth in the table above, Seller shall not be required to open more than two (2) New Centers in any twelve (12) month period. Seller shall use commercially reasonable efforts to obtain all Regulatory Approvals no later than twelve (12) months after a New Center's Opening Date.

Seller shall have the right, in its sole discretion, to open the New Centers within a shorter timeframe. Notwithstanding anything to the contrary in this Agreement, Buyer may reject a Development Proposal, in its sole discretion, for any proposed New Center if such proposed New Center is located in the Mountain Time Zone.

In addition, if there are delays beyond the reasonable control of the Seller, including without limitation, delays in receiving approvals from Buyer regarding the Development Proposal or

obtaining of Regulatory Approvals, or if there is force majeure as set forth in Section 13, or delays due to Landlord, the Opening Date and Regulatory Approval Date shall also be delayed and Seller shall not be liable to Buyer for any damages as a result thereof.

2.4 Operation of the ViroPharma Centers. Seller shall operate the ViroPharma Centers in accordance with current industry standards and the standards it uses in its BPC Centers. In addition, Seller shall comply with the operating covenants for the ViroPharma Centers specified in this Agreement, including sourcing, collecting, testing, using, handling, selling, storing and distributing Plasma in accordance with all Applicable Laws and Regulatory Approvals. The ViroPharma Centers shall be constructed and operated in a manner that is compliant with all Regulatory Approvals.

2.5 Acquisition of Initial and New Centers.

- 2.5.1 Buyer has the option to purchase (or authorize an Affiliate to purchase, provided however that if an Affiliate has the option to purchase, the Buyer will remain primarily liable) all of the Initial Centers (including all related Initial Center Assets), subject to terms and conditions of the Agreement, from Seller between December 31, 2014 and December 31, 2016, upon twelve (12) months prior written notice from Buyer to Seller (for example, Buyer may exercise its purchase option on December 31, 2015, if Buyer notifies Seller by December 31, 2014) provided, however, subject to Section 2.5.5, that the option will expire if Buyer does not provide notice by December 31, 2015, for Initial Centers with a closing date on or before December 31, 2016.
- **2.5.2** Buyer has the option to purchase (or authorize an Affiliate to purchase, provided however that if an Affiliate has the option to purchase, the Buyer will remain primarily liable) all of the New Centers (including all related New Center Assets), subject to terms and conditions of the Agreement, from Seller on or after December 2015 upon twelve (12) months prior written notice from Buyer to Seller (for example, Buyer may exercise its purchase option on December 31, 2016, if Buyer notifies Seller by December 31, 2015) provided, however, subject to Section 2.5.5, that the option will expire if Buyer does not provide notice by December 31, 2016, for New Centers with a closing date on or before December 31, 2017.
- **2.5.3** Notwithstanding the foregoing, if Buyer wishes to exercise its option to purchase it may elect to purchase the New Centers (as set forth in Section 2.5.2) or all of the ViroPharma Centers (as set forth in Sections 2.5.1 and 2.5.2). The Buyer may not elect to purchase only the Initial Centers. If Buyer elects the option to purchase the New Centers only, the Plasma Volume range in Section 4.3 shall be reduced by 140,000 160,000 liters. For the sake of clarity, if Buyer elects the option to purchase the New Centers only, its option to purchase the Initial Centers shall remain in effect as set forth in this Agreement.
- **2.5.4** In addition, the closing dates of the Initial and the New Centers shall be at least twelve (12) months apart.
- 2.5.5 If Buyer notifies Seller of its intent to exercise its Purchase Option, Buyer and Seller must complete the transaction and transfer the Centers within twelve (12) months, unless as set forth above. The Purchase Option shall be exercised in accordance with the terms

of this Agreement and the Purchase Agreement shall be negotiated and executed by the Seller and Buyer (or the Buyer's Affiliate) in accordance with Section 5.1. If Buyer has not provided notice of its intent to purchase the ViroPharma Centers from Seller by the end of the Initial Term or any Renewal Term, and the Agreement is renewed pursuant to Section 9.1, then the Option Period shall be automatically extended to the end of the Renewal Term.

2.5.6 Semi-Annual Meetings. Seller and Buyer shall meet in good faith at least semi-annually to discuss the ViroPharma Centers, the Plasma Volume (as defined below) and any other issues of mutual interest.

Section 3. Financial Matters.

3.1 Obligations of Seller. At all times prior to acquisition of the ViroPharma Centers by Buyer (in which case Buyer shall thereafter be responsible for the costs of operation of each acquired ViroPharma Center), Seller shall be solely responsible for all costs, fees and expenses associated with the cost of construction, operation and maintenance of the ViroPharma Centers.

3.2 Buyer Payments.

3.2.1 Initial Centers

(a) The unused portion of the pre-payments for the Initial Centers will be credited toward the Purchase Price for Plasma in accordance with Section 3.4. As of July 31, 2012, the unused portion of such pre-payments is equal to Two Million, Eight Hundred Twenty-Nine Thousand, Four Hundred Twenty Dollars (\$2,829,420.00) (the "Outstanding Pre-Payments").

3.2.2 New Centers

- (a) Buyer will pay the New Centers' Development Payment of Two Million Dollars (\$2,000,000) as follows:
- (i) One Million Dollars (\$1,000,000) to Seller within fifteen (15) days of the execution of this Agreement, and
- (ii) One Million Dollars (\$1,000,000) to Seller within fifteen (15) days after Seller notifies Buyer that Seller has entered into a lease for the third New Center.
 - (b) In addition, Buyer will pay the following to Seller:
- (i) One (1) Plasma Pre-Payment of One Million Dollars (\$1,000,000) within fifteen (15) days of execution of this Agreement, as a pre-payment on future Plasma purchases, and
- (ii) Four (4) Center Pre-Payments of Seven Hundred Fifty Thousand Dollars (\$750,000) upon the execution of each lease agreement for each New Center, as a refundable pre-payment on future Plasma purchases, for a total of Three Million Dollars

(\$3,000,000). The latter four Center Pre-Payments are each due within fifteen (15) days after Seller notifies Buyer that Seller has entered into a lease for a particular New Center. The Total Pre-Payments will be credited against the Purchase Price for Plasma in accordance with Section 3.4.

TOTAL NEW CENTER PAYMENTS

| Payment Due Date | Amount | Refundable/Non-Refundable | |
|--|-------------|---|--|
| Within fifteen (15) days of Execution of this Agreement | \$2 Million | \$1 Million refundable/\$1 Million non-refundable | |
| Within fifteen (15) days after Seller notifies Buyer that Seller has entered into the 1 st New Center Lease Agreement | \$750,000 | refundable | |
| Within fifteen (15) days after Seller notifies Buyer that Seller has entered into the 2nd New Center Lease Agreement | \$750,000 | refundable | |
| Within fifteen (15) days after Seller notifies Buyer that Seller has entered into the 3rd New Center Lease Agreement | \$1,750,000 | \$1 Million non-refundable/ \$750,000 refundable | |
| Within fifteen (15) days after Seller notifies Buyer that Seller has entered into the 4th New Center Lease Agreement | \$750,000 | refundable | |

3.3 Purchase Price of Plasma Sales.

- (a) Subject to Section 3.4, the Purchase Price of the Plasma purchased by Buyer from the Seller pursuant to Section 4 of the Agreement for 2012 shall be One Hundred Fifty-Eight Dollars and Fifty Cents (\$158.50) per liter of Plasma.
- (b) Beginning January 1, 2013, the Purchase Price shall be increased automatically at the beginning of each calendar year by CPI. In addition, Seller or Buyer may propose an increase or decrease in the Purchase Price of no more than five percent (5%) based on current market conditions for the following calendar year. Such latter increase or decrease shall take effect only if approved by the other Party (such approval shall not be unreasonably withheld). Notwithstanding the above, at no time shall the Purchase Price be less than Seller's Costs plus ten percent (10%).
- (c) Seller shall invoice Buyer for all Plasma at the time of delivery of such Plasma to Buyer. Buyer shall pay undisputed invoices within thirty (30) days from date of Seller's invoice. Such invoice shall not be dated or submitted prior to delivery of the Plasma. All payments shall be made in U.S. Dollars. Seller's invoices shall reflect the actual quantity of

Plasma shipped and the price thereof, as computed in accordance with this Agreement.

(d) In the event of a change to the Specifications, new governmental regulations or FDA or EU required testing, which cause an increase or decrease in Seller's Costs, the Parties shall work in good faith to re-negotiate the Purchase Price.

3.4 Refund of Pre-Payments.

- (a) Buyer shall be entitled to a credit of Six Dollars and Sixty-Seven Cents (\$6.67) per liter of Plasma delivered to Buyer from January 1, 2012 through December 31, 2012 for the Outstanding Pre-Payments.
- (b) Buyer shall be entitled to a credit of Thirteen Dollars and Thirty-Three Cents (\$13.33) per liter of Plasma delivered to Buyer from January 1, 2013 until such time as the aggregate amount of credits (including the credits in Section 3.4(a)) equal the amount of the Outstanding Pre-Payments.
- (c) Buyer shall be entitled to a credit of (i) Six Dollars and Sixty-Seven Cents (\$6.67) per liter of Plasma delivered to Buyer from January 1, 2014 through December 31, 2014 for annual Plasma Volumes above 130,000 liters and (ii) Thirteen Dollars and Thirty-Three Cents (\$13.33) per liter of Plasma delivered to Buyer on or after January 1, 2015 for annual Plasma Volumes above 130,000 liters until such time as the aggregate amount of such credits equals the Refundable Pre-Payments made by Buyer to Seller.
- (d) In the event that this Agreement expires or is terminated for any reason (except a termination of this Agreement by Seller pursuant to Section 9.2) at any time prior to Buyer's full utilization of the credits for the Total Pre-Payments and Buyer does not elect to exercise its Purchase Option, Seller shall, within fifteen (15) days, pay to Buyer an amount equal to any remaining amount of such Total Pre-Payments which has not been so credited to Buyer.

Section 4. Purchase and Sale of Plasma.

- **4.1 Plasma Volume for 2012.** Seller will deliver and Buyer will purchase 70,000 liters of Plasma for the calendar year 2012. The Parties will negotiate in good faith a mutually agreeable and reasonable delivery schedule.
- 4.2 Plasma Volume for 2013. Seller will deliver and Buyer will purchase 170,000 liters of Plasma for the calendar year 2013. The Parties will negotiate in good faith a mutually agreeable and reasonable delivery schedule.
- 4.3 Plasma Volume for 2014. Beginning in 2014, Seller shall sell and Buyer shall purchase the Plasma quantities in a range of 250,000 to 280,000 liters of Plasma on a calendar year basis. Not later than nine (9) months prior to January 1st of a calendar year, Buyer will provide notification of the Plasma Volume required for that calendar year. The Parties will negotiate in good faith a mutually agreeable and reasonable delivery schedule. Seller shall provide the Plasma Volume regardless of the Plasma output at the New Centers.

- **4.4** In the event that Buyer and Seller consummate the purchase of the Initial Centers, Buyer and Seller will meet to adjust the Plasma quantities.
- 4.5 In the event that Buyer and Seller consummate the purchase of all of the ViroPharma Centers, then Buyer shall not be required to purchase any Plasma pursuant to this Agreement, once the Buyer owns all of the ViroPharma Centers.
- 4.6 Plasma Source. Seller will supply and Buyer shall purchase Plasma collected at the Qualified Centers and/or Buyer Approved Collection Centers only, subject to the terms and conditions of this Agreement. During the period of time that any New Center has not received all Regulatory Approvals and is therefore a Non-Qualified Center, Seller shall be solely responsible for any plasma produced at such Non-Qualified Center. Under no circumstance is Buyer required to purchase plasma from any Non-Qualified Center(s) or Buyer Approved Collection Center that has not received all Regulatory Approvals. However, for avoidance for doubt, and for clarification purposes once a New Center has received all Regulatory Approvals, any Plasma collected during the time the New Center was a Non-Qualified Center may be sold to Buyer pursuant to the terms and conditions of this Agreement if and only if such Plasma has been licensed by the FDA and approved by the EU. For the purpose of this Agreement, a center is a "Buyer Approved Collection Center" if it is listed on Exhibit A. Buyer may add a collection center to the list if: (i) the center has received and maintains as current all necessary regulatory approvals, registrations and permits including, but not limited to, FDA licensure, EU approval, iQPP, CLIA, and required state licensing; (ii) the center has completed and Buyer has approved an affirmation of compliance, or the center has been audited by Buyer or its agents pursuant to Section 4.19 with no significant observations; (iii) the center satisfies the requirements described in the "Origin of Plasma" section of the attached Specifications; and (iv) the center has been added to the list of Buyer Approved Collection Centers by Buyer, as set forth in Exhibit A. If any Buyer Approved Collection Center (i) receives a significant enforcement correspondence from a governmental authority (ii) is identified by Buyer or its agents to have significant compliance or quality concerns during an audit under Section 4.19, (iii) or Buyer determines that the plasma collection center is materially deficient in its compliance with applicable laws, regulations or quality requirement, Seller will have thirty (30) business days to provide, in writing, a corrective action plan acceptable to Buyer. If the action plan is found unacceptable or if the Buyer Approved Collection Center cannot provide Plasma within ninety (90) days of any such event, then Buyer may modify Exhibit A to eliminate such Buyer Approved Collection Center from the Seller will supply Plasma to Buyer from the Qualified Centers and Buyer Approved Collection Centers only.
- 4.7 Right of First Refusal of Seller. If Buyer requires, for a certain calendar year, any Plasma in addition to (i) the then-current Plasma Volume supplied under this Agreement plus (ii) 250,000 liters, Buyer must first request an increase in the Plasma Volume up to the applicable maximum Plasma Volume set forth in Sections 4.1 through 4.3 under the terms and conditions of this Agreement. If Seller agrees to provide the additional Plasma Volume in the requested timeframe, then the Plasma Volume will increase by the proposed amount and Buyer will purchase the additional Plasma from Seller in the requested time frame subject to the terms and conditions of this Agreement. If Seller does not agree to provide the additional Plasma Volume within the requested timeframe, then, notwithstanding anything to the contrary in this Agreement, Buyer is permitted to purchase such additional Plasma from an Alternative Supplier.

If the requirement exceeds (i) the maximum Plasma Volume set forth in Sections 4.1 through 4.3 plus (ii) 250,000 liters, then the Buyer shall provide Seller with a right of first refusal to supply such additional Plasma. Buyer will notify Seller in writing of any terms proposed by an Alternative Supplier and Seller will have ten (10) business days to exercise its right of first refusal. If Seller elects not to exercise its right of first refusal or fails to respond to Buyer's request or refuses to supply the requested additional Plasma within such ten (10) business day period to increase the Plasma supply, then Buyer may purchase Plasma from the Alternative Supplier for the time period set forth in the proposal submitted to Seller. For the sake of clarity, Seller may only exercise its right of first refusal to match the terms of the Alternative Supplier's proposal in its entirety. Seller's failure to exercise its option at any time does not waive Buyer's requirement to provide Seller with a right of first refusal for all future plasma needs during the Term of the Agreement.

- 4.8 Hyperimmune Plasma Supply. So long as Seller is able to satisfy all requirements for Plasma pursuant to this Agreement, Seller shall have the right to produce hyperimmune plasma in any New Center for Seller's own use provided that (i) the average volume of hyperimmune plasma does not exceed twenty percent (20%) of the total volume, including Plasma and hyperimmune plasma, produced all of the ViroPharma Centers at any time and (ii) the production thereof in such ViroPharma Centers is in compliance with all applicable Regulatory Approvals and Applicable Laws. Buyer shall not be required to purchase any hyperimmune plasma nor continue any hyperimmune plasma program if Buyer purchases the ViroPharma Centers.
- **4.9 Delivery**. Except as otherwise agreed in writing by the Parties, all deliveries of Plasma shall be FOB designated plasma collection center. Buyer will arrange for a shipping agent or carrier to pick up the Plasma at Seller's facility. The shipping agent or carrier shall be selected and contracted by Buyer.
- 4.10 Governing Documents. All sales of Plasma hereunder shall be subject solely to the terms and provisions of this Agreement (including all exhibits and schedules) and shall not be subject to other terms, conditions or provisions contained in any other purchase order, writings, or documents except to the extent a purchase order, writing or document sets forth or confirms quantity or schedule or place for delivery. Furthermore, in the event of any inconsistency or discrepancy between the terms and conditions of this Agreement and the schedules hereto, or any other record, the terms of this Agreement shall prevail.
- 4.11 Records and Compliance Matters. Seller shall, at its expense, keep and maintain detailed records pertaining to the amount and type of Plasma sold hereunder during the Term and for a period equal to the longer of: (a) thirty (30) years following the date of termination or expiration of this Agreement; and (b) the period of time required by the Applicable Laws. Such records shall be made available for inspection by Buyer during normal business hours, upon reasonable advance written notice, which shall not be less thirty (30) days (or such shorter period of time as may be mutually agreed upon by the Parties). Seller shall transfer such records related to the ViroPharma Centers to Buyer in connection with the consummation of the acquisition of the ViroPharma Centers by Buyer as contemplated by this Agreement.

- 4.12 Licenses and Approvals. Seller shall use commercially reasonable efforts to obtain and maintain in full force and effect all necessary licenses, permits, certifications, consents, registrations, listings, certificates of origin, and any other requisite documents and authorizations, including approvals and registrations, and pay all applicable fees, charges, customs duties and taxes incurred in the performance of its obligations under this Agreement, including with respect to all ViroPharma Centers and Buyer Approved Collection Centers. Without limiting the generality of the foregoing, Seller shall use commercially reasonable efforts to comply with all Applicable Laws, Regulatory Approvals, regulations, rules, and guidelines pertaining to its performances under this Agreement, including but not limited to those set forth in the U.S. Code of Federal Regulations, 21 C.F.R. parts 600-640 and any other applicable local, state or federal law, regulation or ordinance.
- 4.13 Storage, Shipping and Handling of Plasma. Plasma shall be stored, handled, packed and shipped by Seller in such a manner as to prevent damage to the Plasma and containers during shipping, consistent with Seller's current practice and otherwise in compliance with all Applicable Laws and Regulatory Approvals, and shall be stored, handled, packed and shipped (including to any holding facility designated by Buyer) subject to such other conditions set forth in the Specifications. No Plasma shall be released by Seller to Buyer pursuant to this Agreement unless and until such Plasma fully complies with the Specifications, Applicable Laws and Regulatory Approvals, and Seller shall be responsible for ensuring compliance therewith.
- 4.14 Quality. Seller shall source, collect, process, store, distribute, test, transport, and otherwise handle Plasma to be sold to Buyer in accordance with this Agreement, at all times in compliance with all Applicable Laws, Regulatory Approvals and the Specifications. Any and all reasonable changes sought by Buyer to the Specifications shall be sent to Seller for review and approval. Buyer shall advise Seller of any and all anticipated reasonable changes to Specifications as soon as practicable so as to provide Seller with as much advance notice as possible. Seller shall have fifteen (15) business days to agree to implement the changes or to respond to Buyer with its reasons for refusing to implement such changes. If Seller agrees to implement such changes, Seller shall implement such changes as promptly as reasonably possible, but in all cases within sixty (60) days after receipt of written notice of such changes. For purposes of clarity, Seller shall not make changes to the process parameters or the Specifications without prior written approval of Buyer (in its reasonable discretion). Seller shall notify Buyer immediately and in any event not later than five (5) business days after: (a) any Existing Center (including without limitation, any Buyer Approved Collection Center) or ViroPharma Center is closed as a result of regulatory sanctions placed on Seller by the FDA (or any foreign equivalent) or any other governmental authority; (b) Seller or any ViroPharma Center or Buyer Approved Collection Center becomes subject to a significant enforcement action from the FDA (or any foreign equivalent) or any other governmental authority; (c) Seller or any ViroPharma Center or Buyer Approved Collection Center fails an audit or inspection conducted by or on behalf of any governmental authority or any accrediting body; or (d) Seller otherwise learns of significant quality concerns that may impact the safety or quality of the Plasma provided to Buyer. In the event that Seller notifies Buyer of any of the foregoing, or Buyer otherwise determines that any ViroPharma Center or Buyer Approved Collection Center is deficient in its compliance with Applicable Laws, Regulatory Approvals or Specifications, the Parties shall work in good faith to minimize overall shortfalls in quantities of Plasma delivered to Buyer, but any such shortfall shall remain the responsibility of Seller.

- Testing and Approval. Seller shall be responsible, at its sole cost and 4.15 expense, for testing Plasma in accordance with FDA and EMEA requirements and the Specifications. Seller shall be responsible for qualifying appropriate testing centers. Seller shall send all Plasma to be provided to Buyer pursuant to this Agreement only to Approved Testing Centers. However, Seller may send Plasma to an alternative laboratory upon the approval of Buyer, which shall not be unreasonably withheld. With respect to each shipment of Plasma to be shipped to Buyer, Seller shall test such Plasma to ensure compliance with the Specifications, Applicable Laws and Regulatory Approvals, and warranties as set forth in Section 8.2. Seller shall include a certificate of analysis and certificate of compliance, as well as all other documentation described in the Specifications, with each shipment of Plasma, disclosing the results of such testing and certifying conformance with the Specifications and other requirements. Buyer may reject any Plasma, after review of the accompanying documentation or upon inspection after receipt by Buyer or Buyer's designee, if such Plasma is not in compliance with the Specifications or any warranty hereunder. Buyer shall promptly inspect (or have inspected) each shipment of Plasma received from Seller and the associated documentation and shall notify Seller in writing in the event that the Buyer intends to rejects such shipment within ten (10) business days after receipt. In the case of Plasma with defects not readily discoverable during inspection upon receipt by Buyer, Buyer shall notify the Seller of such defects promptly following Buyer's discovery thereof and shall be entitled to reject such Plasma. Rejected Plasma will be destroyed in compliance with Buyer's internal procedures and local environmental requirements. Seller reserves the right to review the relevant records concerning the destruction of rejected batches.
- Remedies For Non-Conformity. In the event that Seller disputes Buyer's 4.16 determination that the Plasma is not in compliance with the Specifications or the warranty hereunder, the Parties shall attempt to cooperate in good faith to resolve the disagreement. If, after thirty (30) calendar days, the Parties are unable to resolve the disagreement, then, at either Party's request, the Parties shall engage an independent testing laboratory or other appropriately qualified expert of recognized repute and credentials, mutually agreeable to the Parties and subject to confidentiality provisions set forth in this Agreement, to analyze a sample of the allegedly nonconforming Plasma and the associated documentation. The laboratory or expert shall use such procedures and tests as such laboratory or expert may consider necessary or appropriate to reach a conclusion. Both Parties agree to cooperate with the independent laboratory's or expert's reasonable requests for assistance in connection with its analysis hereunder. Both Parties shall be bound by the laboratory's or expert's results of analysis, which, absent manifest error, shall be deemed final as to any dispute over nonconformity. The costs incurred by the laboratory or expert shall be borne by the losing Party, or, if the laboratory or expert cannot place the fault noticed and complained about, then the Parties shall share equally the expenses in connection with such laboratory or expert. If the Plasma is determined to be non-conforming, whether by agreement of the Parties or by an independent laboratory, then the Seller will determine whether to either replace the non-conforming Plasma or issue a credit to the Buyer.
- 4.17 Remedies For Supply Failure. Without limiting any other right or remedy Buyer may have hereunder or under law or equity and notwithstanding anything to the contrary in this Agreement, including without limitation Section 4.7, in the event that Seller does not supply the quantities of Plasma as set forth in this Agreement, and Seller does not cure the ability

to supply the Plasma at the quantities requested, within ninety (90) days of notice by Buyer that it will procure such Plasma from an Alternative Supplier, Buyer shall have the right to engage an Alternative Supplier to collect and supply those quantities of Plasma that Seller cannot supply. In such event, Seller shall reimburse Buyer for the following: (i) if the price charged by the Alternative Supplier for such Plasma is higher than the price charged by the Seller, for the difference between such prices, provided Buyer used commercially reasonable efforts to secure the best price from the Alternative Supplier; and (ii) any other reasonable additional incremental costs or expenses arising from Buyer's engagement of the Alternative Supplier of the Plasma. If Buyer is required to enter into a supply agreement with the Alternative Supplier to collect and supply all or part of Buyer's requirements for Plasma then Buyer shall not be required to purchase Plasma pursuant to the terms of this Agreement until the supply agreement with the Alternative Supplier expires. Upon expiration of such supply agreement with the Alternative Supplier, Buyer is required to purchase Plasma from Seller pursuant to the terms of this Agreement (provided Seller is able to supply the Plasma).

- 4.18 Right to Audit. Buyer (or its agent) will have the right to audit Seller's BPC Centers (including without limitation, any Buyer Approved Collection Center) and the ViroPharma Centers and other facilities, systems and records during the Term as they relate to the Plasma provided hereunder to assess Seller's compliance with the Applicable Laws, Regulatory Approvals, Specifications under the terms of this Agreement at Buyer's sole cost and expense, during normal business hours and after providing Seller with reasonable advance notice. Buyer shall coordinate with Seller to schedule audits at mutually agreeable dates and times. Except as set forth below, in no event shall such audits occur more frequently (and with respect to Seller's collection and/or testing centers, per site) than once every twelve (12) months. The limitation on the number of audits per twelve (12) month period shall not apply in the event that (i) a site receives a significant enforcement correspondence, (ii) Buyer identifies a significant compliance or quality concern at a site, or (iii) Buyer otherwise determines that a site is materially deficient in it compliance with applicable laws, regulations or quality requirements, as described in more detail in Section 4.6. No audit of any one collection/and or testing center shall continue in duration for more than two (2) business days. Notwithstanding anything to the contrary in Section 4.11, Seller shall make its records available for inspection by Buyer (or its agent) during any audit pursuant to this Section 4.18. Seller shall respond in writing to all audit observations to Buyer within thirty (30) days of receipt of the audit report. Responses are to include timelines and plans for closure of all corrective actions and commitments.
- 4.19 Regulatory Inspections and Correspondence. Seller shall notify Buyer and, if applicable, provide Buyer with copies of any notices or communications, if any Plasma Center operated by Seller is closed as a result of regulatory sanctions placed or issued by the FDA or equivalent foreign regulatory body, or if any such facility receives a warning letter or consent decree from the FDA or equivalent regulatory body, fails an audit conducted on behalf of any foreign regulatory body or any accrediting body, or is otherwise the subject of similar quality concerns. Seller must notify Buyer in writing immediately and in any event not later than five (5) business days after the earlier of Seller's receipt of the relevant communication or the date Seller learns of such quality concern.
- 4.20 Product Complaints and Adverse Event Reporting. For purposes of clarity, Buyer shall be responsible for receiving all Product complaints and Product adverse

event reports and all communications with complainants. If Seller receives a Product complaint or Product adverse event report, Seller shall refer the complainant to Buyer and provide notification of such complaint to Buyer, including a description of the complaint or event and the name and contact information of the complainant, within three (3) business days after receipt by Seller. Seller shall maintain Biological Deviation Reports ("BDRs") relating to the Plasma and other documentation as required by the applicable regulatory requirements. Seller shall cooperate with Buyer in investigating any Product complaint, Product adverse event or BDR that relates to the Plasma provided hereunder.

- 4.21 Product Regulatory Inspections and Correspondence. For purposes of clarity, Buyer shall be solely responsible for all contacts and communications with the FDA (or any foreign equivalent) and other governmental authorities with respect to all matters relating to the Products. Seller shall forward to Buyer any notices or communications received by Seller concerning the Products immediately, but in no case later than three (3) business days after receipt by Seller.
- 4.22 Plasma Recalls. Each Party will immediately notify the other if either Party discovers any issue that could potentially lead to a recall of the Plasma provided by Seller hereunder. The Parties shall then discuss reasonably and in good faith whether a Plasma recall is appropriate or required, and the manner in which any such recall should be conducted. Seller shall be responsible, at its sole cost and expense, for conducting, and shall have the final decision for, all Plasma recalls after consultation with Buyer (provided that, for clarity, Buyer shall have the sole right to determine whether to recall any Product). The Parties shall cooperate in good faith in coordinating the implementation of any Plasma recall.

Section 5. Purchase of New Centers.

- 5.1 Acquisition. In the event Buyer exercises its Purchase Option under Section 2.5 to require a sale and purchase of the ViroPharma Centers, then the Parties shall negotiate and execute a Purchase Agreement as set forth in Section 5.2 and shall use commercially reasonable efforts to promptly satisfy any and all closing conditions agreed upon in such Purchase Agreement and within the timelines set forth in the Purchase Agreement. The terms and conditions for the closing of such transactions shall be as defined in the Purchase Agreement.
- 5.2 Purchase Agreement. The obligations of Buyer to acquire the ViroPharma Centers and Seller to sell ViroPharma Centers are subject in all respects to the terms of the definitive Purchase Agreement. The Purchase Agreement shall describe the specific assets to be purchased and/or assigned and the liabilities to be assumed and any other rights and obligations of the Parties arising out of such transaction.
- 5.3 Timing of Acquisitions. Once Buyer exercises its Purchase Option to purchase the Initial or New Centers, the Parties will, subject to satisfactory diligence by Buyer, to be performed within thirty (30) days of the Purchase Option, or the Buyer is deemed to have acquiesced, act in good faith and use their commercially reasonable efforts to negotiate, draft and execute the Purchase Agreement within six (6) months from the date that Buyer notifies Seller of its election to exercise the Purchase Option under Section 2.5.

5.4 Acquisition Cost of ViroPharma Centers. The purchase price for each ViroPharma Center (including the related ViroPharma Center Assets) shall be equal to the amount set forth below, <u>less</u> any remaining amounts of the Total Pre-Payments that have not been refunded to Buyer hereunder and shall be paid no later than fifteen (15) days after the execution of a Purchase Agreement:

| Year of Purchase | Purchase Price per Qualified Center | Purchase Price per Non- Qualified Center |
|------------------|--|---|
| 2012 | \$3,700,000 | \$2,700,000 |
| 2013 | \$3,800,000 (*) | \$2,800,000 (*) |
| 2014 | \$3,900,000 (*) | \$2,900,000 (*) |
| 2015 | \$4,000,000 (*) | \$3,000,000 (*) |
| 2016 | 2015 Purchase Price multiplied by CPI | 2015 Purchase Price multiplied by CPI |
| 2017 | 2016 Purchase Price multiplied by CPI | 2016 Purchase Price multiplied by CPI |

^(*)Beginning January 1, 2013, the Purchase Price shall be increased automatically at the beginning of each calendar year by CPI

5.5 Additional Acquisition Cost of Non-Qualified Centers. The purchase price for each Non-Qualified Center (including the related ViroPharma Center Assets) to be purchased shall be equal to the amount set forth above in Section 5.4, less any remaining amounts of the Total Pre-Payments that have not been refunded to Buyer; provided however, that Seller will be responsible for obtaining all necessary Regulatory Approvals for each Non-Qualified Center so that the center becomes a Qualified Center within twelve (12) months from such Non-Qualified Center's Opening Date, subject to reasonable delays as set forth in the last paragraph in Section 2.3. If Seller successfully obtains the Regulatory Approvals for Non-Qualified Centers as set forth in the previous sentence, then Buyer shall pay to the Seller, within fifteen (15) days of Buyer's receipt of notification of such Regulatory Approvals from Seller, an additional One Million Dollars (\$1,000,000) to the Seller for each Qualified Center. Beginning January 1, 2013, the additional One Million Dollars (\$1,000,000) in the preceding sentence shall be increased automatically at the beginning of each calendar year by CPI.

5.6 ViroPharma Center Assets.

(a) Pursuant to the Purchase Agreement, Seller shall use commercially reasonable efforts to sell, assign, transfer, convey and deliver to the Buyer (or the Buyer's Affiliate), and the Buyer (or the Buyer's Affiliate) shall purchase, acquire and accept from the Seller, all of the Seller's right, title and interest in and to the ViroPharma Centers and all assets

related to the ViroPharma Center, in each case free and clear of any liens and any other encumbrances, unless otherwise specified in the Purchase Agreement, including the following ViroPharma Center Assets, as shall be more specifically expressed in the definitive Purchase Agreement: (i) those certain parcels of real property (including all buildings, improvements and structures located thereon and all appurtenances thereto) owned or leased by Seller on which the ViroPharma Centers are situated; (ii) all of the leasehold interests and rights of the Seller, including such subleases under the tenant space leases, ground leases and other leases of real property; (iii) all fixed assets, fixtures, furnishings, furniture, office supplies, tools, machinery and equipment owned or leased by the Seller and used in the operation of the ViroPharma Centers; (iv) right to offer employment to all employees employed exclusively at the ViroPharma Centers, (v) all inventories of general production supplies used in the operation of the ViroPharma Centers, with the exception of Hyperimmune supplies; (vi) a perpetual, nonexclusive, royalty free license to use the SOPs for any lawful purpose, licenses, permits, contracts, agreements, arrangements and/or commitments listed in the Purchase Agreement, until such time as Buyer obtains their own licenses, permits, contracts, agreements, arrangements and/or commitments; (vii) all business and financial records and personnel and donor records relating exclusively to the ViroPharma Centers; (viii) copies of the Seller's proprietary data bases, donor lists and records, donor center technical guides, quality control and training manuals, specialty guides; and (ix) all Regulatory Approvals related to such ViroPharma Centers; (x) all good will of the Seller related to the ViroPharma Centers; and (xi) any other items as may be reasonably required for Buyer to operate the ViroPharma Center.

- (b) The Purchase Agreement shall also provide that the Buyer (or Buyer's Affiliate) shall assume and shall thereafter pay, discharge and perform in the ordinary course solely those obligations arising after the closing date under the Purchase Agreement and as identified therein as "Assumed Obligations." The Buyer (or Buyer's Affiliate or designee) shall not assume and shall not be liable for any liabilities or obligations of the Seller other than as expressly identified in the Purchase Agreement as an Assumed Obligation. The Seller shall remain responsible for all liabilities and obligations related to the ViroPharma Center arising or accrued prior to the closing date of the transactions contemplated by this Section unless specified in the Purchase Agreement.
- (c) The Purchase Agreement shall contain provisions that are normal and customary in such an agreement, including but not limited to the normal representations and warranties, indemnifications, covenants, conditions precedent to closing, etc.
- (d) To the extent that any ViroPharma Center Assets are unable to be transferred to the Buyer (or the Buyer's Affiliate) at closing due to regulatory requirements, or contractual restrictions, the Parties will negotiate in good faith, a mutually agreeable transition services agreement at closing, which will require Seller to continue to operate the ViroPharma Centers under its licenses, permits, etc., and provide all required assistance to Buyer to complete the required regulatory process, until the completion of the required regulatory process and the transfer of such ViroPharma Center Assets to Buyer, which services shall be covered by the management fee set forth in Section 5.10.
- **5.7 Default**. In the event that a Party defaults under this Section, the other Party hereby reserves all rights and remedies in law and at equity, including such Party's right to sue

for damages and specific performance.

5.8 Change of Control of Seller.

- (a) If, during the Term, Seller signs a definitive agreement regarding a Proposed Transaction, then Seller shall notify Buyer in writing within five (5) business days of such Proposed Transaction. Within thirty (30) days after receipt of notice by Buyer of such Proposed Transaction, Buyer shall have the right (but not the obligation) to exercise its Purchase Option to acquire the ViroPharma Centers, regardless of whether such election is during the Option Period or not, at the price specified in Sections 5.4 and 5.5 above and Seller has no further obligations to the ViroPharma Centers. Notwithstanding anything to the contrary in this Agreement, if Buyer elects to exercise its Purchase Option due to a Proposed Transaction pursuant to this Section 5.8, then (i) the twelve (12) months written notice from Buyer to Seller under Section 2.5 shall not be required, (ii) Buyer shall purchase all of the ViroPharma Centers at once and the twelve (12) months waiting period set forth in Section 2.5.4 shall not apply, and (iii) the Parties shall work in good faith to negotiate and execute the Purchase Agreement as soon as possible but in no event later than the closing date of the Proposed Transaction.
- (b) Notwithstanding anything to the contrary in Section 5.10 or elsewhere in this Agreement, in the event that Buyer purchases the ViroPharma Centers pursuant to this Section 5.8, Buyer shall not be required to enter into the Management Agreement with Seller and, in its sole discretion, may elect to (i) enter into the Management Agreement with Seller pursuant to Section 5.10, (ii) enter into a management agreement with a third party, or (iii) manage the acquired ViroPharma Centers itself.
- 5.9 Change of Control of Buyer. If, during the Term, Buyer enters into a definitive agreement regarding a Proposed Transaction, then Buyer shall notify Seller in writing within five (5) business days of such Proposed Transaction. Seller shall have the right (but not the obligation) to terminate the Purchase Option set forth in Section 2.5 by notifying Buyer in writing within thirty (30) days after receipt of notice by Buyer of such Proposed Transaction.
- Management of ViroPharma Centers. In the event that Buyer elects to exercise its Purchase Option for the ViroPharma Centers in accordance with the terms of this Agreement, the Buyer and Seller shall (unless Buyer elects not to enter in the Management Agreement in its sole discretion), within six (6) months from the notice of Buyer's election to exercise its Purchase Option, negotiate in good faith and enter into the Management Agreement, whereby Seller would provide management services at the ViroPharma Centers similar to the services allocated to Seller's plasma operations and general and administrative costs at the BPC If, after good faith negotiations, the Parties cannot reasonably agree on the Management Agreement within such six (6) month period, the Parties shall not be required to enter into the Management Agreement. In addition to any other terms and conditions contained in such Management Agreement, unless otherwise agreed to by the Parties, such Management Agreement shall include provisions that Seller shall be entitled to a yearly management fee in an amount equal to thirty dollars (\$30) per liter of Plasma produced by the ViroPharma Centers, plus an annual increase in the management fee, based on the increase in CPI for each annual period, beginning January 1, 2011. The initial term of the Management Agreement will be three (3) years. The Management Agreement may be renewed by either Party for two (2) year renewal

periods, provided, however, that such Party gives the other Party written notice of its intent to renew at least eighteen (18) months prior to the end of the initial term or any renewal term and the other party agrees to renew the Management Agreement.

Section 6. Construction of New Centers.

- 6.1 Description of Services. Seller shall perform (or subcontract for) all engineering and construction services (hereinafter referred to as the "Construction Services") for each of the ViroPharma Centers, prior to the execution of the Purchase Agreement. The Construction Services will be performed to normal, customary and reasonable standards.
- 6.2 General Terms. Seller is and shall operate as an independent contractor in regards to this Agreement and not as an agent or employee of Buyer. Seller shall use commercially reasonable efforts to ensure that all subcontracts, supply agreements, lease agreements, rental agreements and other agreements entered into regarding the ViroPharma Centers contain a provision allowing Seller to freely assign such agreement and that Seller is released of any liability under the lease agreement, provided however, that the requirement that Seller can freely assign a lease agreement, shall be subject to the condition that Buyer signs an assignment and assumption of lease agreement and agrees to disclose reasonable financial information requested by a landlord. Further, Buyer is obligated to provide lease guaranties if necessary.

Section 7. Operational Matters Regarding ViroPharma Centers

- 7.1 Affirmative Covenants. With respect to the operations of the ViroPharma Centers and Buyer Approved Collection Centers, the Seller and its Affiliates will:
- (a) At all times, operate the ViroPharma Centers and Buyer Approved Collection Centers in a reasonable and prudent manner in compliance with all Applicable Laws, its FDA-approved and EU-approved Standard Operating Procedures ("SOP") and all other standards generally practiced in the industry, and shall also construct (with respect to the ViroPharma Centers) and operate the ViroPharma Centers and Buyer Approved Collection Centers in a manner that is compliant with all Regulatory Approvals, including Foreign Regulatory Approvals, and qualify for approval by the relevant regulatory authorities in the United States and the European Union. Consistent with the foregoing, the Seller shall use its reasonable efforts consistent with good business practice to preserve the goodwill of the suppliers, contractors, licensors, employees, customers, distributors and others having business relations with the Seller.
- (b) Pay and discharge all lawful taxes, assessments and governmental charges or levies imposed upon it, upon its income and profits or upon any of its assets, before the same shall become in default, as well as all lawful claims for labor, materials and supplies which, if unpaid, might become a lien or charge upon such properties or any part thereof.
- (c) Do or cause to be done all things necessary to preserve and keep in full force and effect its corporate existence, rights and franchises and to comply in all respects with all laws, regulations and orders of each governmental authority having jurisdiction over the ViroPharma Centers or Buyer Approved Collection Centers.

- (d) Use commercially reasonable efforts to maintain management personnel with substantially the same qualifications and experience as the management personnel at its BPC Centers.
- (e) Use commercially reasonable efforts to maintain, preserve, protect and keep its property, including all ViroPharma Center Assets (and the assets of or related to the Buyer Approved Collection Centers), in good repair, working order and condition and will, from time to time, make all necessary and reasonable repairs, renewals, replacements, betterments and improvements thereto.
- (f) Keep adequately insured, by financially sound reputable insurers, all ViroPharma Center Assets (and the assets related to the Buyer Approved Collection Centers) and other property of a character insured the same as the BPC Centers.
 - (g) Be the sole owner of the ViroPharma Centers.
- **7.2 Negative Covenants**. The Seller shall not, with respect to the ViroPharma Centers or ViroPharma Center Assets:
- (a) make any material change in its operations which differ from its BPC Centers, except such changes as may be required to comply with any applicable requirements of law;
- (b) sell, lease (as lessor), transfer or otherwise dispose of, or mortgage or pledge, or impose or suffer to be imposed any encumbrance on, any ViroPharma Center or any ViroPharma Center Assets other than in the ordinary course of business or if approved by Buyer, which shall not be unreasonably withheld;
- (c) enter into any contracts of sale, options to purchase or rights of first refusal (except as set forth in this Agreement) with respect to the ViroPharma Centers during the Option Period;
- (d) institute any material increase in any profit-sharing, bonus, incentive, deferred compensation, insurance, pension, retirement, medical, hospital, disability, welfare or other employee benefit plan with respect to the employees of the ViroPharma Centers that differs from any plan offered to the Seller's employees at the BPC Centers;
- (e) make any material change in the compensation of the employees of the ViroPharma Centers, that differs from Seller's employees compensation at the BPC Centers;
- (f) otherwise engage in any practice, take any action, or enter into any transaction which would cause a Material Adverse Effect.

Section 8. Warranties.

8.1 Warranties of Seller Regarding Construction of ViroPharma Centers. Seller will, for the protection of Buyer, request from all vendors and contractors from which Seller procures machinery, equipment or materials or services, warranties and guarantees with

respect to such machinery, equipment, materials or services, which shall be made available to Buyer to the full extent of the terms thereof. Without limiting the generality of the foregoing, Seller shall render all commercially reasonable assistance to Buyer for the purpose of enforcing the same, without incurring any cost or liability.

8.2 Warranties of Seller Regarding Operation of ViroPharma Centers and the Sale of Plasma.

- (a) Seller represents and warrants that Seller shall use commercially reasonable efforts to ensure that the Plasma delivered hereunder shall conform to the Specifications set forth in this Agreement and shall be collected, processed, tested, stored, handled and delivered to Buyer in accordance with the Specifications, Applicable Laws and Regulatory Approvals. Seller further represents and warrants that it has conducted all donor selection, screening, eligibility, consent and testing activities for the Plasma delivered hereunder in accordance with Applicable Laws, rules, and regulations. Seller further warrants and represents that it shall use its commercially reasonable efforts to ensure the Plasma delivered hereunder shall not, as of the date of delivery to or placement with Buyer's carrier, be adulterated or misbranded with the meaning of the Federal Food, Drug and Cosmetic Act, shall be in compliance within the Biological Products sections of the Public Health Service Act and applicable regulations and shall be in compliance with any applicable international, federal, state or local laws or regulations.
- (b) Seller warrants and represents that clear and unrestricted title for all Plasma purchased under this Agreement will pass to the Buyer upon acceptance by Buyer as defined in Section 4.
- (c) Seller represents and warrants that Seller has, or will use it commercially reasonable efforts to obtain, all applicable Regulatory Approvals, permits and licenses required in the performance of its obligations under this Agreement, including without limitation the FDA and the EU requirements. Seller certifies it will not use in any capacity the services of any person, including any firm or individual that has been debarred or, to the best of its knowledge, is subject to debarment under the Generic Drug Enforcement Act of 1992, amending the Food, Drug, and Cosmetic Act of 21 U.S.C. 335a (a) or (b). Seller agrees to notify Buyer promptly in the event any person providing services to Seller under the scope of this Agreement is debarred or, to the best of its knowledge, becomes subject to debarment.
- (d) Seller shall maintain (and cause to be maintained) such records as are necessary and appropriate to demonstrate compliance in the collection, processing, testing, , transport, storage, disposal and other handling of Plasma with applicable Regulatory Approvals and Applicable Laws.
- (e) Unless otherwise mutually agreed by the Parties, all Plasma shipped to Buyer shall have been drawn within the eighteen (18) month period prior to the date of receipt of such Plasma by Buyer or its applicable designee.

8.3 General Representations and Warranties.

8.3.1 Seller's Representations and Warranties.

- (a) The Seller is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. The Seller has all requisite corporate power and authority to execute, deliver and perform this Agreement and all other agreements entered into or delivered in connection with the transactions contemplated hereby. The Seller is qualified to do business as a foreign corporation in each location where a ViroPharma center will be opened. The Seller has or will use commercially reasonable efforts to obtain all authorizations, approvals, orders, licenses, certificates and permits of and from all governmental or regulatory bodies necessary to own and/or lease the properties and assets employed by the Seller in the conduct of operating a plasma collection center at the ViroPharma Centers and BPC Centers and to conduct its business and operations as currently conducted.
- The execution, delivery and performance of this Agreement and all other agreements entered into in connection with the transactions contemplated hereby have been duly and validly authorized by all necessary corporate action on the part of the Seller. This Agreement has been duly executed and delivered by the Seller, constitutes the valid and binding obligation of the Seller, and is enforceable in accordance with its terms. All other agreements to be entered into pursuant to this Agreement by the Seller in connection with the transactions contemplated hereby will be duly executed and delivered by the Seller, will constitute the valid and binding obligations of the Seller, and will be enforceable in accordance with their respective terms. The execution, delivery and performance of this Agreement does not, and all other agreements entered into in connection with the transactions contemplated hereby by the Seller will not, violate, conflict with, result in a breach of or constitute a default under (or an event which with due notice or lapse of time, or both, would constitute a breach of or default under) or result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, creation of any lien, security interest or other encumbrance under (a) the Certificate of Incorporation or By-laws of the Seller, as amended to date, (b) any note, agreement, contract, license, instrument, lease or other obligation to which the Seller is a party or by which it is bound or to which any of its assets are subject, (c) any judgment, order, decree, ruling or injunction or (d) any statute, law, regulation or rule of any governmental agency or authority.
- (c) There is no action, lawsuit, proceeding, claim, controversy, arbitration or investigation pending or, to the Seller's knowledge, threatened against, or directly involving, the Seller's plasma collection business, inclusive of the ViroPharma Center Assets. There is no unsatisfied or outstanding order, writ, judgment, injunction or decree affecting the Seller's plasma collection business or the ViroPharma Center Assets. The Seller has complied and is complying with all laws, ordinances, and governmental rules and regulations applicable to it and its properties, assets and business, and has obtained or will use commercially reasonable efforts to obtain all Regulatory Approvals necessary for the ownership of its properties and the conduct of its business as currently conducted.
- (d) Except for obtaining the Regulatory Approvals for ViroPharma Centers and the landlord's consent, Seller is unaware of any other consent or approval of any third party or governmental body that is required for the consummation by the Seller of the transactions contemplated by this Agreement.
- (e) Seller has not made and, during the Term, will not make any commitments to, or grant of any rights to, any other Person that is or may be inconsistent or in

conflict with any rights granted to Buyer under this Agreement, unless both Parties mutually agree in writing.

8.3.2 Buyer's Representations and Warranties.

- (a) The Buyer is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. The Buyer has all requisite power and authority to execute, deliver and perform this Agreement and all other agreements entered into or delivered in connection with the transactions contemplated hereby.
- The execution, delivery, and performance of this Agreement and all other agreements entered into in connection with the transactions contemplated hereby have been duly and validly authorized by all necessary corporate action on the party of the Buyer. This Agreement has been duly executed and delivered by the Buyer, constitutes the valid and binding obligation of the Buyer and is enforceable against it in accordance with its terms. All other agreements to be entered into pursuant to this Agreement by the Buyer in connection with the transactions contemplated hereby will be duly executed and delivered by the Buyer, will constitute the valid and binding obligations of the Buyer, and will be enforceable in accordance with their respective terms. The execution, delivery and performance of this Agreement does not, and all other agreements to be entered into in connection with the transactions contemplated hereby by the Buyer will not, violate, conflict with, result in a breach of or constitute a default under (or an event which with due notice or lapse of time or both, would constitute a breach of or default under) or result in the creation of any lien, security interest or other encumbrance under (a) its charter or By-laws, (b) any note, agreement, contract, license, instrument, lease or other obligation to which the Buyer is a party or by which it is bound, (c) any judgment, order, decree, ruling or injunction or (d) any statute, law, regulation or rule of any governmental agency or authority.

Section 9. Term and Termination.

- 9.1 Term of Agreement. The term of this Agreement shall commence on the Effective Date and shall continue in full force and effect until the end of the Initial Term, unless earlier terminated in accordance with the terms contained herein. The Agreement may be renewed for additional Renewal Term(s), provided either Party notifies the other in writing at least eighteen (18) months prior to the end of the Initial Term or any given Renewal Term and the other party agrees to renew the Agreement.
- 9.2 Termination for Cause. Either Party shall have the right to immediately terminate this Agreement in the event the other Party fails to perform any of its material obligations under this Agreement and such failure to perform is not cured within sixty (60) days of written notice of such failure, prior to the purchase of the Initial Centers; provided however that Buyer's monetary obligations must be cured within fifteen (15) days after written notice of late payment. The right of any Party to terminate this Agreement pursuant to this Section shall not be affected in any way by its waiver or failure to take action with respect to any prior default. The Party not in default shall be entitled to terminate this Agreement without prejudice to any other rights conferred on it by this Agreement or under law or equity. A termination shall not relieve a Party from any obligations that survive termination or expiration of this Agreement. If

Buyer terminates the Agreement pursuant to this Section 9.2, Seller shall refund to Buyer within fifteen (15) days an amount equal to one hundred thirty percent (130%) of the Total Pre-Payments which have not been refunded to Buyer pursuant to Section 3.4 as of the date of termination. If Seller terminates the Agreement pursuant to this Section 9.2, Seller is not required to refund the Total Pre-Payments which have not been refunded to Buyer pursuant to Section 3.4 as of the date of termination and Seller shall use commercially reasonable efforts to obtain an alternative purchaser. Buyer shall reimburse Seller to recover the difference between (i) Plasma purchase price to such alternative purchaser and the then current Plasma price in effect under this Agreement, if the price under this Agreement is higher and (ii) any other reasonable additional incremental costs or expenses arising from Seller's engagement of the alternative purchaser of the Plasma. If Seller cannot obtain an alternative purchaser than Buyer must pay to Seller the Purchase Price.

- 9.3 Other Termination Provisions. Either Party may immediately terminate this Agreement, and have no further liability under this Agreement, if the other Party: (i) starts a proceeding, or indicates its acquiescence to a proceeding started by another, relating to it under any bankruptcy, reorganization, rearrangement, insolvency, readjustment or debt, dissolution, liquidation or similar law; (ii) makes an assignment for the benefit of creditors; (iii) consents to the appointment of a receiver, trustee or liquidator for a substantial part of its property; (iv) files, or has filed against it, a petition in bankruptcy, reorganization, rearrangement or insolvency which, if filed against it, is not dissolved or dismissed within ninety (90) days after filing; or (v) had entered against it an order by a court of competent jurisdiction appointing a receiver, trustee or liquidator for it or a substantial part of its property, or approving its dissolution or termination, and if not consented to or acquiesced in by such Party, such order in not vacated or set aside or stayed within ninety (90) days.
- Agreement, all further obligations by the Parties with respect to the purchase and sale of Plasma (including, for clarity, any Plasma Volume obligations imposed on Buyer), as well as the construction, operation and sale of the ViroPharma Centers shall immediately expire, and the Parties shall have no further obligation to proceed with any transactions with respect to such matters (except pursuant to any management agreement executed in connection with the purchase of a ViroPharma Center or any Purchase Agreement which has not been consummated to the extent Buyer and Seller elect to continue such transaction. The rights and remedies available to Buyer or the Seller are as set forth in this Agreement and shall not preclude or dismiss Buyer's or the Seller's right to pursue any other or additional right or remedy, including, without limitation, any claim for damages against any third party in connection with this Agreement. The failure to exercise any right or remedy in the event of any breach or default shall not constitute a waiver or adversely affect Buyer's or the Seller's right to exercise any right or remedy under this Agreement.
- 9.5 Provisions Surviving Termination. Termination or expiration of this Agreement shall not relieve the Parties hereto of any obligation accruing prior to such termination or expiration. The provisions of this Agreement which by their nature would continue beyond any termination or expiration of this Agreement, including Section 8, Section 9, Section 10, Section 11, Section 12, Section 13, Section 14 and Section 15 shall survive any termination or expiration of this Agreement to the degree necessary to permit their complete

fulfillment or discharge.

9.6 Termination of Prior Agreements. The Parties hereby agree that, upon the execution of this Agreement, all prior agreements for the purchase and sale of Plasma between Buyer and Seller are hereby terminated in all respects, other than any provisions of any such agreement whose terms specifically survive termination; provided, however that this Section 9.6 shall not apply to (i) the Plasma Purchase/Sale Agreement entered into as of May 15, 2009 by and between Seller and Buyer, and (ii) the Plasma Purchase/Sale Agreement entered into as of October 8, 2009 by and between Seller and Buyer. This Agreement does not affect any agreements between Biotest AG and Buyer or Buyer's Affiliate(s).

Section 10. Indemnity.

- 10.1 Indemnification. The Seller and Buyer hereby indemnify and agree to hold harmless each other and its respective affiliates, agents, employees, officers and directors, from and against any and all claims, losses, liabilities, damages, attorney's fees, costs and expenses which may be sustained by and/or claimed against the other Party by virtue of the gross negligent performance of services rendered by the other Party, the willful misconduct by the other Party or its officers, employees or agents, or any representation or warranty contained in this Agreement being breached, untrue or materially misleading by omission or otherwise. It being understood, however, that the financial liability under this section shall be limited to the extent of each Party's insurance coverage, if such coverage is in effect and in accordance with any requirements under this Agreement at the time a claim is asserted under this section.
- 10.2 Defense and Settlement. The obligations to indemnify, defend and hold harmless set forth in this section shall not apply to the Party to be indemnified (the "Indemnified Party") unless the Indemnified Party (i) notifies the Party providing such indemnification (the "Indemnifying Party") as soon as practicable of any matters in respect of which the indemnity may apply and of which the Indemnified Party has knowledge; (ii) gives the Indemnifying Party, at the Indemnifying Party's option, the full opportunity to control the response thereto and the defense thereof, including any agreement relating to the settlement thereof, provided that the Indemnifying Party shall not settle any such claim or action without the prior written consent of the Indemnified Party (which consent shall not be unreasonably withheld or delayed) or such settlement include as an unconditional term thereof the giving by the claimant of an unconditional release from all liability in favor of the Indemnified Party; and (iii) cooperates with the Indemnifying Party, at the Indemnifying Party's cost and expense, in the defense or settlement thereof. Notwithstanding the foregoing, the indemnification obligations hereunder shall not be relieved hereunder for failure to do the foregoing, or delay with so doing, unless the Indemnifying Party is materially prejudiced thereby. In addition, the Indemnified Party may, at its own expense, participate in its defense if any claim.

Section 11. Limitation of Liability.

11.1 EXCEPT AS PROVIDED IN THIS AGREEMENT, NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY OR ANY OTHER PERSON OR ENTITY FOR ANY SPECIAL, INCIDENTAL, CONSEQUENTIAL, INDIRECT, PUNITIVE OR EXEMPLARY DAMAGES HOWEVER CAUSED (INCLUDING ARISING OUT OF OR

RELATED TO THIS AGREEMENT, THE PERFORMANCE OF THE SERVICES), REGARDLESS OF THE FORM OF ACTION, WHETHER FOR BREACH OF CONTRACT, BREACH OF WARRANTY, TORT, NEGLIGENCE, STRICT PRODUCT LIABILITY, OR OTHERWISE (INCLUDING, WITHOUT LIMITATION, DAMAGES BASED ON WILLFULNESS, LOSS OF PROFITS, LOST REVENUES, OR LOSS OF BUSINESS OPPORTUNITY), AND WHETHER OR NOT SUCH PARTY HAS BEEN ADVISED OR KNEW OF THE POSSIBILITY OF SUCH DAMAGES.

- THE FOREGOING LIMITATION OF LIABILITY SHALL NOT APPLY TO (A) A MATERIAL BREACH BY A PARTY OF ITS CONFIDENTIALITY OBLIGATIONS PURSUANT TO THIS AGREEMENT, OR (B) DAMAGES, LOSSES AND CLAIMS ARISING OUT OF A PARTY'S GROSS NEGLIGENCE, WILLFUL MISCONDUCT OR FRAUD, OR (C) A PARTY'S INDEMNIFICATION OBLIGATIONS UNDER THIS AGREEMENT ARISING FROM THIRD PARTY CLAIMS. THE PARTIES ACKNOWLEDGE THAT THIS LIMITATION OF LIABILITY PROVISION HAS BEEN SEPARATELY NEGOTIATED, IS A MATERIAL INDUCEMENT TO THE PARTIES ENTERING INTO THIS AGREEMENT ON THE TERMS PROVIDED HEREIN AND SHALL BE ENFORCEABLE REGARDLESS OF WHETHER ANY REMEDY PROVIDED FOR FAILS OF ITS ESSENTIAL PURPOSE.
- ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT, FROM ANY AND ALL CAUSES, WHETHER BASED ON CONTRACT, TORT (INCLUDING NEGLIGENCE), OR ANY OTHER CAUSE OF ACTION EXCEED THE PURCHASE PRICE FOR THE PLASMA AS DETERMINED IN ACCORDANCE WITH SECTION 3.3 OR IF BUYER EXERCISES ITS PURCHASE OPTION FOR A GIVEN VIROPHARMA CENTER, THE PURCHASE PRICE OF A VIROPHARMA CENTER AS DETERMINED IN ACCORDANCE WITH THE FORMULA SET FORTH IN SECTION 5.4, AS APPLICABLE.

Section 12. Confidentiality.

During the Term and for a period of ten (10) years thereafter, each Party shall (i) hold the other Party's Confidential Information in strict trust and confidence and avoid the disclosure or release thereof to any other person or entity by using at least the same degree of care as it uses to avoid unauthorized use, disclosure, or dissemination of its own Confidential Information of a similar nature, but not less than reasonable care, (ii) not use the other Party's Confidential Information for any purpose whatsoever except as expressly contemplated under this Agreement, and (iii) not directly or indirectly, copy, reproduce, use, publish, misappropriate, assign, or otherwise transfer or disclose to any person the other Party's Confidential Information, other than as permitted pursuant to the terms of this Agreement, regardless of whether such information was actually delivered to the receiving Party prior to the effective date of this Agreement. "Confidential Information," as used herein, shall mean all information, trade secrets, inventions, data, processes, or other records relating to a Party's business, financial affairs or operations, including but not limited to information related to business plans, technology, source code, product or service requirements, customers, pricing, techniques and methods, which is either marked or identified as confidential or which the receiving Party knew or reasonably should have known, under the circumstances, was confidential, whether disclosed in any

tangible, electronic, visual or other medium. Confidential Information shall also include all information, know-how, trade secrets, data (technical or non-technical) or other confidential information concerning the operations, projects, organization, business or finances of a Party or any third party to which a Party owes a duty of confidentiality, in whatever form, that a receiving Party learns, generates or acquires in conjunction with the performance of the services pursuant to this Agreement. All files, records, documents, notes, or other items relating to or embodying Confidential Information that may be delivered to a receiving Party or to which a receiving Party may be granted access, shall remain the exclusive property of the Party disclosing such Confidential Information. Neither Party shall disclose to any third parties, including a customer of the other Party, the existence or terms of this Agreement. For purposes of clarity, all information relating to any ViroPharma Center shall be deemed to be the Confidential Information of Buyer from and after the consummation of the acquisition of such ViroPharma Center by Buyer. On either Party's request upon the termination or expiration of this Agreement, the other Party shall immediately: (i) stop using the Confidential Information of the requesting Party; (ii) return all materials provided by the requesting Party that contain Confidential Information of the requesting Party, except for one copy that may be retained by the other Party's legal counsel to confirm compliance with the obligations under this Agreement; (iii) destroy all copies of Confidential Information of the requesting Party in any form, including materials prepared by or for the other Party and Confidential Information contained in computer memory or data storage apparatus or materials prepared by or for Seller; and (iv) provide a written confirmation to the requesting Party that such other Party has taken all the foregoing actions.

12.2 A receiving Party shall not disclose any Confidential Information except to its officers, directors, employees and authorized representatives, and its Affiliates and their respective officers, directors, employees and authorized representatives (collectively "Representatives") who need to know such information for the purpose of performing the services contemplated by this Agreement and which persons shall be similarly bound in writing (it being understood that each such Representatives shall be informed by the receiving Party of the confidential nature of such material and shall be directed to treat such material confidentially in accordance with the terms of this Agreement). A receiving Party agrees to be responsible for any breach of this Agreement by any of its Representatives. Notwithstanding the foregoing, a receiving Party shall not be required to maintain confidentiality with respect to information (i) which is or becomes part of the public domain not due to any act or omission by the receiving Party; (ii) of which it had independent knowledge without confidentiality restriction prior to disclosure by the disclosing Party, as shown by written evidence; ; (iii) which comes into the possession of the receiving Party (without confidentiality restriction) in the normal and routine course of its own business from and through independent, non-confidential sources, as shown by written evidence;; or (iv) which is required to be disclosed by receiving Party governmental requirements. If receiving Party is requested or required (by oral questions, interrogatories, requests for information or document subpoenas, civil investigative demands, or similar process) to disclose any Confidential Information supplied to it by the disclosing Party, the receiving Party shall, if possible, immediately notify the disclosing Party of such request(s) prior to any disclosure so that the disclosing Party may seek an appropriate protective order. acknowledges that Buyer may be required to (and shall be permitted to) disclose the terms of this Agreement pursuant to its reporting obligations under the Securities Exchange Act of 1934, as amended (or other securities exchange requirements), and to file a copy of this Agreement as an exhibit to a periodic, quarterly or annual report required to be filed by Buyer thereunder.

- 12.3 Failure on the part of the receiving Party to abide by this Section 12 may cause the disclosing Party irreparable harm for which damages will not be an adequate remedy at law. Accordingly, the non-breaching Party has the right to seek injunctive relief to prevent any threatened or actual violations of this section in addition to whatever remedies it may have at law, In such a proceeding, the Party allegedly breaching this Agreement expressly waives the defense that a remedy in damages will be adequate and any requirement in an action for specific performance or injunction for the posting of a bond by the non-breaching Party.
- Section 13. Force Majeure. Neither Party will be liable for any default or delay in the performance of its obligations under this Agreement if and to the extent such default or delay is caused, directly or indirectly, by fire, flood, elements of nature or other acts of God, epidemics, any outbreak or escalation of hostilities, war, terrorism, riots or civil disorders, strikes or work stoppage, utility or telecommunications failures or fluctuations, proclamation, regulation, or ordinance or other act to order of any court, government or governmental agency or any other similar cause beyond the reasonable control of such party, except that each Party shall be responsible for the timely payment of all of its financial obligations to the other Party. In any such event, the non-performing Party will be excused from any further performance and observance of the obligations so affected only for as long as such circumstances prevail and as long as such Party continues to use commercially reasonable efforts to recommence performance or observance as soon as practicable.

Section 14. Additional Agreements Between the Parties.

- 14.1 Notifications. The Buyer, on the one hand, and the Seller, on the other hand, shall promptly notify the other of any action, suit or proceeding that shall be instituted or threatened against such Party to restrain, prohibit or otherwise challenge the legality of any transaction contemplated by this Agreement. Each Party will give prompt written notice to the other Party of any material adverse development causing a breach of any of its own representations and warranties in this Agreement. No disclosure by any Party pursuant to this Section, however, shall be deemed to prevent or cure any misrepresentation, breach of warranty, or breach of covenant.
- 14.2 Non-Competition. During the Term of this Agreement and for a period of five (5) years after any termination or expiration of this Agreement or the Management Agreement, whichever is longer, except for a termination based on the breach of this Agreement by Buyer, Seller agrees that it will not directly enter into or become associated with or otherwise operate any plasma collection centers within a radius of ten (10) miles from the location of any ViroPharma Center.
- 14.3 Non-Solicitation. During the Term, and the term of the Management Agreement, and for two (2) years thereafter, the Parties shall not, directly or indirectly hire or attempt to hire any employee of the other Party who performed substantial work on any project covered by this Agreement or the Management Agreement (including any employees at the ViroPharma Centers which are acquired by Buyer hereunder) without the other Party's prior written consent; provided that the foregoing shall not prohibit Buyer from hiring the employees

employed exclusively at the ViroPharma Centers pursuant to its Purchase Option.

- 14.4 Obtaining Permits and Licenses. The Seller shall be responsible for obtaining all Regulatory Approvals required by any governmental agency with respect to the construction and operation of the ViroPharma Centers and the Buyer Approved Collection Centers, prior to the purchase of any ViroPharma Center. The Seller will use its reasonable efforts and act diligently to secure any consents and approvals, including the Regulatory Approvals, required to effect the transactions contemplated by this Agreement. The Buyer agrees to cooperate and participate to affect the transfer of permits as necessary or to obtain permits that cannot be transferred.
- 14.5 Insurance. Seller and Buyer shall provide and maintain in full force and effect, usual and customary insurance coverage relating to or arising under this Agreement, including, errors and omissions, products liability, general liability and related insurance coverage with policy limits with at least the following minimums:
 - (a) Commercial General Liability coverage of \$5,000,000.00 per incident and \$5,000,000.00 in aggregate.
 - (b) Products and Completed Operations Liability coverage of \$5,000,000.00 per incident and \$5,000,000.00 in the aggregate.
 - (c) Workers compensation as required by federal, state and local law, if applicable.
 - (d) Employers Liability limits of \$1,000,000.00 per incidents, if applicable.
- every other person working on its behalf, has not and will not, in connection with the transactions contemplated by this Agreement or in connection with any other business transactions involving Buyer, make, offer or promise to make any payment or transfer anything of value, directly or indirectly, (i) to any governmental official or employee (including employees of government-owned and government-controlled corporations and public international organization), (ii) to any healthcare professional or organization, (iii) to any political party, official of a political party or candidate, (iv) to an intermediary for payment to any of the foregoing, or (v) to any other person or entity if such payment or transfer would violate the laws of the country in which made or the laws of the United States. It is the intent of the parties that no payments or transfers of value shall be made which have the purpose or effect of public or commercial bribery, acceptance of or acquiescence in extortion, kickbacks or other unlawful or improper means of obtaining business. This section shall not, however, prohibit normal and customary business entertainment or the giving of business mementos of nominal value.

Section 15. General.

(a) Press Releases and Public Announcements. Neither the Buyer, on the one hand, nor the Seller, on the other hand, shall, without the approval of the other, make any press release or other public announcement concerning the transactions contemplated by this Agreement, except as and to the extent that any such Party shall be so obligated by law, in which

case the other Party shall be advised and the Parties shall use their commercially reasonable efforts to cause a mutually agreeable release or announcement to be issued; provided, however, that the foregoing shall not preclude communications or disclosures necessary to implement the provisions of this Agreement or to comply with the accounting and disclosure obligations of the Securities and Exchange Commission or the rules of any stock exchange or NASDAQ or to enable the Buyer or Seller to obtain debt or equity financing.

- (b) **No Third-Party Beneficiaries**. This Agreement shall not confer any rights or remedies upon any Person other than the Parties and their respective successors and permitted assigns.
- (c) Entire Agreement. This Agreement (including the exhibits and schedules hereto and the documents referred to herein) constitutes the entire agreement between the Parties and supersedes any prior understandings, agreements, or representations by or between the Parties, written or oral, to the extent they relate in any way to the subject matter hereof, including any prior agreements between the Parties; provided, however that this Section 15(c) shall not apply to (i) the Plasma Purchase/Sale Agreement entered into as of May 15, 2009 by and between Seller and Buyer, and (ii) the Plasma Purchase/Sale Agreement entered into as of October 8, 2009 by and between Seller and Buyer.
- (d) Succession and Assignment. This Agreement shall be binding upon and inure to the benefit of the Parties named herein and their respective successors and permitted assigns. No Party may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other Party, not to be unreasonably withheld; provided, however, that Buyer or Seller may, without the consent of the other Party, but upon written notice (i) assign any or all of its rights and interests hereunder to one or more of its Affiliates; or (ii) designate one or more of its Affiliates to perform its obligations hereunder (in any or all of which cases Buyer nonetheless shall remain responsible for the performance of all of its obligations hereunder).
- (e) **Counterparts**. This Agreement may be executed in one or more counterparts (including by means of facsimile), each of which shall be deemed an original but all of which together will constitute one and the same instrument.
- (f) **Notices**. All notices, requests, demands, claims, and other communications hereunder shall be in writing. Any notice, request, demand, claim, or other communication hereunder shall be deemed duly given (i) when delivered personally to the recipient, (ii) one (1) business day after being sent to the recipient by reputable overnight courier service (charges prepaid), with delivery confirmation, (iii) one (1) business day after being sent to the recipient by facsimile transmission or electronic mail, confirmed with an error-free transmission report, addressed as set forth below. or (iv) four (4) business days after being mailed to the recipient by certified or registered mail, return receipt requested and postage prepaid, and addressed to the intended recipient as set forth below:

If to Buyer:

If to Seller:

ViroPharma Biologics, Inc.

Biotest Pharmaceuticals Corporation 5800 Park of Commerce Boulevard, N.W.

c/o ViroPharma Incorporated 730 Stockton Drive Exton, PA 19341 Attn: Chief Operating Officer

Facsimile: (610) 458-7380

With a copy (which shall not constitute notice) to:

ViroPharma Incorporated 730 Stockton Drive Exton, PA 19341 Attn: General Counsel Facsimile: (610) 458-7380 Boca Raton, FL 33487 Attn: Chief Executive Officer Facsimile: (561) 989-5890

With a copy (which shall not constitute notice) to:

Biotest Pharmaceuticals Corporation 5800 Park of Commerce Boulevard, NW Boca Raton, FL 33487 Attn: Legal Department Facsimile: (561) 989-5517

Any Party may change the address to which notices, requests, demands, claims, and other communications hereunder are to be delivered by giving the other Party notice in the manner herein set forth.

- (g) Governing Law. This Agreement shall be governed by and construed in accordance with the domestic laws of the State of Delaware without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other, jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware.
- (h) Amendments and Waivers. No amendment of any provision of this Agreement (inclusive of any exhibits or schedules) shall be valid unless the same shall be in writing and signed by Buyer and Seller. No waiver by any Party of any provision of the Agreement or any default, misrepresentation, or breach of warranty or covenant hereunder, whether intentional or not, shall be valid unless the same shall be in writing and signed by the Party making such waiver nor shall such waiver be deemed to extend to any prior or subsequent default, misrepresentation, or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence. The failure of any Party hereto to enforce at any time any provision of this Agreement shall not be construed to be a waiver of such provision, nor in any way to affect the validity of this Agreement or any part hereto or the right of any Party thereafter to enforce each and every such provision.
- (i) **Severability**. Wherever possible, each provision hereof shall interpreted in such manner as to be effective and valid under applicable law, but in case any one or more of the provisions contained herein shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, such provision shall be ineffective to the extent, but only to the extent, of such invalidity, illegality or unenforceability without invalidating the remainder of such invalid, illegal or unenforceable provision or provisions or any other provision hereof, unless such a construction would be unreasonable.
- (j) **Expenses**. Each of Buyer and Seller will bear its own costs and expenses (including legal fees and expenses) incurred in connection with this Agreement and the transactions contemplated hereby, except as otherwise specified in this Agreement.

- (k) Interpretation. Any reference to any federal, state, local, or foreign statute or law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise. The word *including* shall mean including without limitation. The exhibits and schedules identified in this Agreement are incorporated herein by reference and made a part hereof. Articles, title and headings to sections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.
- (l) **No Strict Construction**. The Parties acknowledge and agree that: (i) each Party and its counsel reviewed and negotiated the terms and provisions of this Agreement and has contributed to its revision; (ii) the rule of construction to the effect that any ambiguities are resolved against the drafting Party shall not be employed in the interpretation of this Agreement; and (iii) the terms and provisions of this Agreement shall be construed fairly as to all Parties hereto and not in favor of or against any Party, regardless of which Party was generally responsible for the preparation of this Agreement.
- (m) **Independent Parties**. This Agreement shall not be deemed to create any partnership, joint venture, amalgamation or agency relationship between Buyer and Seller. Each Party shall act hereunder as an independent contractor.
- (n) Further Assurances. Each Party shall execute and deliver such additional instruments and other documents and use commercially reasonable efforts to take or cause to be taken, all actions and to do, or cause to be done, all things necessary under Applicable Law to consummate the transactions contemplated hereby.

Remainder of page intentionally left blank. Signature page follows.

IN WITNESS WHEREOF, the duly authorized representatives of the Parties have executed this Agreement as of the date first above written.

BUYER:

SELLER:

VIROPHARMA BIOLOGICS, INC.

BIOTEST PHARMACEUTICALS CORPORATION

/s/ Daniel B. Soland

/s/ Jordan Siegel

By: Daniel B. Soland

By: Jordan Siegel

Title: President

Title: Chief Executive Officer

Schedule 1

VIROPHARMA INCORPORATED

SOURCE PLASMA SPECIFICATIONS

A. Origin of Plasma

- 1. Source Plasma must be collected from donors at USA-FDA licensed plasma center(s) and in accordance to all Applicable Laws and any relevant FDA Memoranda and guidances.
- 2. Source Plasma must be collected in accordance to current applicable EU regulations and the European Pharmacopoeia Monograph for "Human Plasma for Fractionation."
- 3. Plasma centers must be IQPP certified and adhere to PPTA voluntary standards. Source Plasma must be from plasma centers in compliance with PPTA IQPP Viral Marker Standards.
- 4. Source Plasma must be processed and stored according to All Applicable laws.
- 5. Copies of FDA registrations and European certificates shall be provided for each plasma center and testing laboratory. Updates to these documents shall be provided on an annual basis.

B. Selection and Exclusion Criteria for Donors and Plasma Donations

- 1. The donor/donation selection shall be in accordance with All Applicable Laws and any relevant FDA guidelines; the current applicable EU regulations and any relevant technical annexes and directives.
- 2. The donor eligibility criteria shall comply with the current FDA regulations and with applicable EU directives.
- 3. Acceptable plasma shall be only from Qualified Donors as defined by IQPP standards.

C. <u>Testing Specifications</u>

- 1. Serology tests on individual units shall be in accordance with FDA requirements and performed with FDA-licensed test kits. Each unit shall be tested and found negative for the following:
 - Hepatitis B Surface Antigen
 - Hepatitis C Antibody
 - HIV-1/HIV-2 Antibody

- 2. Nucleic Acid Amplification tests (NAT) to be performed in accordance to FDA requirements (mini pools) and found negative for:
 - HCV RNA
 - HIV-1 RNA
 - HBV DNA (performed with FDA licensed test kit or under an IND).
- 3. Additional NAT tests to be performed in accordance to established USA and European guidelines:
 - HAV RNA
 - Parvovirus B19 DNA
- 4. Each donor must be tested and found negative in accordance to USA and European requirements for:
 - Syphilis Antibody
 - Atypical Antibody
- 5. Any additional tests or modifications of existing tests shall be implemented as soon as they are required by the FDA and/or the European Pharmacopoeia monograph.
- 6. A list of all current test kits or methods shall be submitted to ViroPharma Biologics, Inc., and updated in the event a test or method is changed.

D. Plasma Collection and Processing Procedures

- 1. Source Plasma collection in bottles in accordance to FDA and European requirements current at the time of donation.
- 2. Procedures must be maintained and followed to ensure all donations are traceable to the individual donors.
- 3. Each Source Plasma bottle will be labeled with at least the following information:
 - Supplier name and license number
 - Plasma Type
 - Unit Identification will be by means of bar-coded unit number CODE 128
 - Volume
 - Anticoagulant composition
 - Storage temperature
 - Expiration date
 - Plasma bottle lot number (on bottle)
- 4. Source Plasma must be frozen by cooling rapidly within 24 hours of collection in validated freezers to ensure that a temperature of -25°C or colder is reached at the core of each plasma unit within 12 hours of placing in the freezer.
- 5. Source Plasma collected in non-EU approved plasma centers must be frozen at -20°C or colder immediately upon collection in accordance to FDA requirements.
- 6. Plasma units that fall into one of the following categories are unacceptable for shipment to ViroPharma Biologics, Inc.:
 - Units with reactive or positive viral marker test results; Prior or subsequent units from donors that have been found to be reactive/positive for required testing

- Hemolyzed or lipemic units
- Units with frozen plasma on the outside of the container
- Broken or contaminated units
- Untested or units with incomplete testing (except for units shipped to RxCrossroads under "holding")
- Applicant/Orphan units
- Units having errors that breach traceability such as units that cannot be traced back to an individual donor

E. Plasma Storage and Transport Requirements

- 1. Source Plasma must be stored and transported at -20°C or colder. All storage and transportation temperatures must be recorded and documented.
- 2. Temperature deviations during storage or shipment must be reported to ViroPharma Biologics, Inc. prior to shipment of the plasma. Salvaged Plasma as defined under 21 CFR is not acceptable to ViroPharma Biologics, Inc.
- 3. Source Plasma shipped to ViroPharma Biologics, Inc. must have at least a remaining shelf life of five (5) years as of delivery date.
- 4. One sample representing each unit must be provided with the shipment to ViroPharma Biologics, Inc. packaged in separate boxes.
- 5. Each packing case will be labeled at least with the following information:
 - Supplier name and license number
 - Plasma Type
 - Shipment Id
 - Case number

F. Lookback and Post-Donation Procedures

- 1. Plasma supplier must comply with all current FDA and European lookback/post donation reporting requirements.
- 2. Lookbacks and notification for destruction of plasma must be made to ViroPharma Biologics, Inc.:
 - Within three (3) working days of receipt of reactive test results from a donor from whom prior or subsequent units have been shipped to ViroPharma Biologics, Inc.
 - Within one (1) working day of notification of Post Donation Information resulting in product recalls or seizure concerning units shipped to ViroPharma Biologics, Inc.
 - Within five (5) working days of receipt of Post Donation Information not resulting in a seizure or recall (e.g. tattoo, body piercing, high risk behavior).

G. Epidemiological Data Reports

1. Each plasma center must comply with PPTA IQPP Viral Marker Standards and EMEA data reporting requirements. Reports must be provided to ViroPharma Biologics, Inc. annually or more frequently upon request.

H. Required Shipment Documentation

- 1. Each shipment of Plasma must have the following shipment documentation and electronic information associated with it.
 - Bill of Lading (copy)
 - Certificate of Quality (original)
 - Shipping Report Summary
 - Shipment Report Detail
 - Plasma Packing and Test Report Forms
 - E-File (see requirements below)
- 2. The Bill of Lading shall include the following information:
 - Unique shipment number (Format Center ID + 4 digit Shipment Number)
 - Bill of Lading #
 - Shipment date
 - Collecting facility's name and address
 - Plasma type
 - Total volume of shipment
 - Total number of units in the shipment
 - Total number of cases in the shipment
 - Temperature statement
- 3. The Shipping Report Summary shall include the following information:
 - Shipment number
 - Collecting facility's name and address
 - License #
 - Plasma type
 - Cases, units and volume by collection week
 - Total cases, units and volume of shipment
 - Earliest and latest bleed date within the shipment
 - Shipment date & time
 - Bill of Lading number
 - Temperature of plasma freezer at time of shipment & signature
- 4. The Shipping Report Detail shall include the following information:
 - Shipment number
 - Shipment date
 - Collecting facility's name and address
 - License number
 - Plasma type
 - Units and volume by case
 - Total cases, units and volume of shipment
- 5. The Ship to Holding Report (when required) shall include the following information:
 - Shipment number
 - Shipment date
 - Testing statement for hold shipments
 - Manager and QA signature and date

- 6. The Plasma Packing and Test Report shall include the following information:
 - Shipment number
 - Plasma type
 - Collection year and week
 - Case number
 - Collecting facility's name and address
 - Donation date
 - Donor number (Donor Number Format: 5 characters numeric; e.g., 12345)
 - Unit number
 - Volume
 - Test results
 - Total units and volume for each case
 - Signature and date of reviewer
- 7. The originals of this documentation are to be mailed to the following address:

Theresa Gwaltney

ViroPharma Biologics, Inc.

1821 Prelude Drive

Vienna, VA 22182

I. Shipment Electronic Data

- 1. Every shipment will include an electronic file that contains information about the shipment. Included in these files is the following information:
 - Either a text-file (ASCII) or a CSV-file be used as a standard.
 - Structured as:
 - o 1 header-record

The header should describe the details of the shipment.

o n detail-records, one per plasma-unit

The detail-records should describe the details of one donation.

- A trailer-line is useful, as it enables the program reading the file to check if all records are available and that the input-file is not corrupt.
- o Use semi-colon as the data separator.
- Header-record containing the following information:
 - o H Indicating that this line is a header-line
 - o Supplier Supplier Name
 - o BOL Bill of Lading #
 - o SNO-number Unique shipment number (Format Center ID + Shipment Number)

Shipment-date Shipment-date (Format ddmmyyyy)

- o File-date Date on which this file was created (Format ddmmyyyy)
- A number of **detail-records** containing the following information for each plasma unit:
 - o D Indicating that this line is a detail-line
 - o Line-number Sequential-number

o BTC

The name of the plasma center (Format: two alpha characters are Center unique ID; – See table in Schedule Exhibit A-1)

o Plasma-type
o Unit-number

Type of plasma delivered (Source)

Donation

number

(Format: 2 character unique Center ID + 7 digit unit number)

o Donation Date
o Volume

Volume

Test results

Test results

- HIV1/HIV2 Test Result
- Anti-HCV Test Result
- HBsAg Test Result
- HCV PCR Test Result
- HIV-1 PCR Test Result
- HAV PCR Test Result (If Required)
- HBV PCR Test Result (If Required)
- PARVOB19 Test Result (If Required)
- Trailer-line containing the following information:
 - o E Indicating that this line is a bottom-line
 - o #Lines Number of lines
 - o <CR><LF> Carriage Return-Line Feed sequence

Exhibit A

Buyer-Approved Collection Centers

| Biotest Plasma Center | 2704 Peach Orchard Road, Augusta, Georgia 30906 |
|-----------------------|--|
| Biotest Plasma Center | 1112 North Main Street, Gainesville, FL 32601 |
| Biotest Plasma Center | 408 S. Gilbert St., Iowa City, IA 52240 |
| Biotest Plasma Center | 300 S. 17 th Street, Lincoln, NE 68508 |
| Biotest Plasma Center | 2301 N. University Drive, #103, Pembroke Pines, FL 33024 |
| Biotest Plasma Center | 2860 Cerrillos Road, Suite B1, Santa Fe, NM 87507 |
| Biotest Plasma Center | 1027 Commerce Boulevard, Dickson City, Pennsylvania 18519 |
| Biotest Plasma Center | 444 Martin Luther King Blvd., Youngstown, OH 44502 |
| Biotest Plasma Center | 233 West Hancock Ave., Athens, GA 30601 |
| Biotest Plasma Center | 2320 Cleveland, Ft. Myers, FL 33901 (planned relocation September 2012 to 4391 Colonial Boulevard, Fort Myers, FL 33966) |
| Biotest Plasma Center | 233-C Western Blvd., Jacksonville, NC 28548 (planned relocation September 2012 to 1213 Country Club Road, Jacksonville, NC 28541-0411) |
| Biotest Plasma Center | 711 Broadway Street, San Antonio, TX 78215 |

Exhibit B

Approved Testing Centers

Serology and Donor Qualification test will be performed by:

Qualtex Laboratory
6211 IH 10 West
San Antonio, Texas 78201

NAT testing shall be performed by:

National Genetics Institute 240 S. Sepulveda Blvd. #235 Los Ángeles, CA 90064

Or

Qualtex Laboratory 6211 IH 10 West San Antonio, Texas 78201

EXHIBIT "C"

INITIAL CENTERS

Biotest Plasma Center 1007 Commerce Blvd. Dickson City, PA 18519

Biotest Plasma Center 2704 Peach Orchard Road Augusta, Georgia 30906

Biotest Plasma Center 233 West Hancock Avenue Athens, Georgia